Author's response to reviews

Title: Survey of HBsAg-positive pregnant women and their infants regarding measures to prevent maternal-infantile transmission

Authors:

Guo Yan (swallow315@126.com)
Meng Leping (34763784@qq.com)
Hu Meina (sanliuhu@163.com)
Du Yukai (duyukai100@yahoo.com.cn)

Version: 5 Date: 22 December 2009

Author's response to reviews:

Dear Editor:

We have provided an “IRB Approval/Informed consent” in the Methods section of the revised manuscript. The contents reviewed by Ethics Committee include, Implementation of programs, the objects and amount, informed consent, etc. If necessary, We can provide scanned IRB Approval. We will also upload the title page.

Thanks very much again for your attention to our paper#Once again, thank you for your help to our paper processing.

Yours sincerely#

Yan Guo, Ph.D.